Author's response to reviews

Title: Unstated factors in orthopaedic decision-making: a qualitative study

Authors:

    Rachael Gooberman-Hill (R.Gooberman-Hill@bristol.ac.uk)
    Anna Sansom (lionbuffalo@yahoo.co.uk)
    Caroline M Sanders (caroline.sanders@manchester.ac.uk)
    Paul A Dieppe (Paul.Dieppe@pms.ac.uk)
    Jeremy Horwood (J.Horwood@bristol.ac.uk)
    Ian D Learmonth (Ian.Learmonth@bristol.ac.uk)
    Susan Williams (s.williams@bristol.ac.uk)
    Jenny L Donovan (jenny.donovan@bristol.ac.uk)

Version: 3 Date: 6 September 2010

Author's response to reviews:

Please find attached the final version of our manuscript. We have addressed all the formatting changed as requested:

Author list is now in journal style
We have renamed the Discussion as 'Discussion and Conclusions'
We have made Table One into a single table rather than as two parts
We have ensured that authors contributions are correct
We have removed track changes
The Introduction has been renamed 'Background'
We have checked for typography and have made typographical edits and changes to improve grammar and clarity where needed.

We would like to thank you for your careful reading of our manuscript.

With regards,

Rachael Gooberman-Hill and co-authors